Author’s response to reviews

Title: Non-medical prescribing in Australasia and the UK: the case of podiatry

Authors:

Alan M Borthwick (ab12@soton.ac.uk)
Anthony J Short (anthony@podiatrypractice.com.au)
Susan A Nancarrow (S.Nancarrow@sheffield.ac.uk)
Rosalie Boyce (rosalie.boyce@gmail.com)

Version: 9 Date: 3 January 2010

Author’s response to reviews:

Revisions required have now been done.

References: References should be cited in the text using consecutive numbers. The paragraph that was added during the last revision has, unfortunately, disrupted the order of the citations. Currently reference 94 is cited before 92 and 93. Please either re-place citations in the correct order, or alter the order of the reference list, so that citations run consecutively.

- The order of the reference list has now been altered to ensure the correct order.

Many thanks

Anthony Short